



Titre: Title:	Impact of public and private research funding on scientific production: The case of nanotechnology
Auteurs: Authors:	Catherine Beaudry et Sedki Allaoui
Date:	2012
Référence: Citation:	Beaudry, Catherine et Allaoui, Sedki (2012). Impact of public and private research funding on scientific production: The case of nanotechnology. <i>Research Policy</i> , <i>41</i> (9), p. 1589-1606. doi:10.1016/j.respol.2012.03.022



Open Access document in PolyPublie

URL de PolyPublie: PolyPublie URL:	http://publications.polymtl.ca/2308/
Version:	Version finale avant publication / Accepted version Révisé par les pairs / Refereed
Conditions d'utilisation: Terms of Use:	CC BY-NC-ND

Document publié chez l'éditeur commercial

Document issued by the commercial publisher

Titre de la revue: Journal Title:	Research Policy
Maison d'édition: Publisher:	Elsevier
URL officiel: Official URL:	http://dx.doi.org/10.1016/j.respol.2012.03.022
Mention légale: Legal notice:	In all cases accepted manuscripts should link to the formal publication via its DOI

Ce fichier a été téléchargé à partir de PolyPublie, le dépôt institutionnel de Polytechnique Montréal

This file has been downloaded from PolyPublie, the institutional repository of Polytechnique Montréal

http://publications.polymtl.ca

Impact of public and private research *funding* on scientific production: The case of nanotechnology

Catherine Beaudry^{ab §}

catherine.beaudry@polymtl.ca

and

Sedki Allaoui^a

allaouisedki@gmail.com

^a Ecole Polytechnique of Montreal P.O. Box. 6079, Downtown office Montreal, Qc, H3C 3A7 Canada

^b Centre for Interuniversity Research on Science and Technology (CIRST) Université du Québec à Montréal P.O. Box 8888, Downtown office Montreal, Qc, H3C 3P8 Canada

Abstract

This article measures the impact of public grants, private contracts and collaboration on the scientific production of Canadian nanotechnology academics. The paper estimates a time-related model of the impact of academic research financing and network structure on the research output of individual academics measured by the number of papers. Results suggest that the effect of individual public funding follows a J-shaped curve. Although contracts have no effects, the impact of patenting follows an inverted-U shaped curve. In addition, a strong central position in the past collaborative network has a positive effect on research output.

Keywords: Scientific articles; Public research funding; Contracts; Patents; Innovation networks; Nanotechnology

[§] Corresponding author: Department of mathematics and industrial engineering, Ecole Polytechnique of Montreal, P.O. Box. 6079, Downtown office, Montreal, Qc, H3C 3A7, Canada, tel. : +1 514 340-4711 extension 3357, fax. : +1 514 340-4173, <u>catherine.beaudry@polymtl.ca</u>

1 Introduction

The financing of university research is being scrutinized in Canada and elsewhere as governments increasingly demand measures of impact and outcomes from grant awarding bodies. One such impact is scientific output: Payne and Siow (2003) as well as Blume-Kohout et al. (2009) for instance show that the public funding of university research has a positive effect on university scientific production. As Gibbons et al. (1994) suggest, the direction of research has migrated from a discipline-based science free of societal needs to a more demand driven science that must meet certain objectives. The integration of university research financing with government programs aimed at the stimulation of innovation is a relatively recent phenomenon (Link and Scott, 2004), but a strategy that is used more and more. This is especially true for domains seen as potential growth generators such as aerospace, biotechnology and lately nanotechnology. Innovation incentives such as R&D tax credits have been put in place and university researchers have benefitted from increased links with industry as a consequence in the form of research contracts.

While public research grants are generally associated with wide scope projects, private contracts concentrate on short-term objectives aiming at the production of knowledge that can rapidly be used (Goldfarb, 2008). Gulbrandsen and Smeby (2005) find a positive correlation between private financing of Norwegian academics and the number of articles they publish. Not only are these university researchers more productive, their work is more applied and they collaborate more with other academics and with industrial researchers. Van Looy et al. (2004) find similar results for contracting activities of researchers.

Both publicly funded research and private contracting activities may lead to patenting. The literature generally finds that academics who patent produce more scientific publications (Breschi et al., 2005; Calderini and Franzoni, 2004; Calderini et al., 2007; Meyer, 2006; Van Looy et al., 2006) or that there is no significant effect (Agrawal and Henderson, 2002). Balconi and Laboranti (2006) show that researchers that are named inventors on patents owned by firms publish more than their colleagues. Although it is not the prime aim of this paper to investigate whether patenting is detrimental to scientific publication performance, our model allows us to crudely contribute to the debate.

As funding does not generally come to single researchers but mainly to teams of researchers, the other issue that this paper wishes to discuss is the influence of collaboration, measured for instance by the position of individual scientists within their co-publication network, on individual scientific production. Lee and Bozeman (2005) find that US researchers who collaborate generally produce more scientific articles than researchers working alone. Other authors find similar results for the UK (Katz and Hicks, 1997) and for Quebec (Landry et al., 1996). International collaboration increases the potential to publish and to publish in international mainstream journals (Barjak and Robinson, 2007; Martín-Sempere et al. 2002). To our knowledge, however, very little or no literature examines the influence of individual indicators of the position of scientists within the collaboration network on scientific production.

This article thus integrates the literature on university research funding and on innovation networks. Following in the footsteps of the literature, our goal is then to estimate the influence of publicly and privately funded research on the production of scientific articles controlling for the collaborative network position of the researchers and taking into account the university affiliation of the scientists. As such, we aim to answer the following questions: Do better-funded

scientists produce more articles? Does contracting and working with industry slow down this productivity? Are better-connected scientists more productive?

The remainder of the paper proceeds as follows: section 2 describes the conceptual framework including a brief review of the literature; section 3 presents the data and methodology; section 4 presents various summary statistics of the data; section 5 presents and analyses the regression results; and finally, section 6 concludes.

2 Conceptual framework

A great number of scientific publications have been published in nanotechnology during the last two decades (see for instance the review by Palmberg et al., 2009, and the country comparison studies by Youtie et al., 2008). The observation of this rapid rate of growth in nanotechnology since 1990 is further supported by Bonaccorsi and Thoma (2007). It is thus a multidisciplinary field that has created a rich pool of publications in all these different disciplines (Porter et al., 2008 and Mogoutov and Kahane, 2007 have developed specific keyword searches for different subdomains).

As such, nanotechnology has potential applications in a wide range of domains, from nanomaterials to nanoelectronics, and from nanofoods to nanobiotechnology. Lipsey et al. (2005) and Youtie et al. (2008) indeed suggest that nanotechnology may be the next general purpose technology, i.e. the next engine of growth of the 21st century.

As an emerging technology, nanotechnology innovation goes through different phases of development. For the first phase, the patience required and the uncertainties linked to the innovation process do not suit all types of fund providers, hence the importance of public research in these domains. Collaboration with university research centers contribute to reducing the uncertainties of innovation activities and thus the risks related to R&D projects. Public financing of research thus facilitates the production of knowledge and constitutes one of the key elements of the development of high-tech innovation. Research funding therefore enhances the success of innovation.

Zucker et al. (2007) insist on the necessity of collaboration between institutions to foster information exchange. Multi-project research centers force researchers and their universities to collaborate more efficiently using the diversity of resources available. Collaboration can indeed become a powerful lever to raise funds (Daniel et al., 2003), and scientific collaboration and research funding are consequently intrinsically intertwined.

Nanotechnology research requires an important infrastructure in order to realize the projects that lead to the development of new products or processes (Robinson et al., 2007), even more so than for biotechnology. Infrastructure investment not only consists of the acquisition of complex machines and instruments but also of the funds necessary for their operation and maintenance that requires specialized skilled personnel and logistics management to optimize access to this infrastructure. The creation of such a technological platform hence requires long-term investment in order to play its role as a space that fosters the emergence and fulfillment of new opportunities. Although universities and their affiliated research centers are the natural candidates for the localization of such technological platforms, because their main goal is often to generate knowledge aimed at yielding technological innovation, the private sector should play an important role in their management and utilization.

Intuitively, most university research projects are financed from public funds (Hagedoorn et al., 2000). Hart (2001) recognizes that public financing is increasingly targeted towards the

development of innovation according to specific political guidelines; the main objective of these political guidelines being the creation and maintenance of competence centers within the university network. This network, composed of universities and their affiliated research centers, is an integral part of the national system of innovation (Hall et al., 2003). Within the national system of innovation, universities and their affiliated research centers play a crucial role in the innovation value chain that leads to economic growth (Link and Scott, 2004; Zucker et al., 2002). For instance, public research infrastructure generates technological knowledge that is transferred via pure externalities, entrepreneurship or contracts (Zucker *et al.*, 1998). University researchers facilitate knowledge transfer between research laboratories and firms (Audretsch and Stephan, 1996). All these beneficial effects have a common origin: money and to a certain extent, collaborative work.

Payne and Siow (2003) as well as Blume-Kohout et al. (2009) suggest that the augmentation in public financing that university science research has experienced recently has had a tremendous effect on scientific output. Furthermore, public funding for specific projects can be perceived as a signal of quality not only for the funded researchers, but also for their university. Adams et al. (2005) show that top universities and departments that have earned awards, have larger teams (with an increased scientific division of labor at the international level) and have more government funding.

The granting of research money further act as a signal that attracts additional funding in subsequent years. Using a static model, Arora et al. (1998) show that the publication track record of researchers has an influence on future grants and consequently on future publication levels as well. Zucker et al. (2007) generally show the major impact that research financing has both on the publication of scientific articles and on patent production. Specifically, Payne and Siow

(2003) show that three years after receiving an instrumentation grant of one million dollars, between 8 and 11 new articles are generally published annually. Jacob and Lefgren (2007) find that specific grants contribute to adding one additional publication within the five years subsequent to the attribution of the grant.

Proposition 1: Academics that obtain greater amounts of public funding generate a larger number of articles.

In a field as close to its science base as nanotechnology, public financing should be seen as complementary to private contracts. Private funding has dramatically increased since 1985 (Cohen *et al.*, 2002) and so did university-industry research centers. University-industry research relations increase contact with scientists but lower communication between scientists and possibly shield their work from potentials future partners (Welsh et al., 2008). As a consequence, a greater proportion of contracts could have a negative effect on a scientist's future scientific production. This could be due to the different reward process used in industrial and academic milieus (Owen-Smith *et al.*, 2001), or to the short-termism associated with private contracts (Goldfarb, 2008). Despite these findings, Blumenthal et al. (1996), Gulbandsen and Smeby (2005) as well as Van Looy et al. (2004) find that contracts have a positive effect on knowledge production. In fact, an overwhelming body of evidence finds a positive relationship between 'academic enterprise' and scientific performance (see the excellent review of Larsen, 2011 for a complete survey).

Proposition 2: Receiving greater amounts of research funds from contracts is beneficial to the scientific productivity of an academic.

Knowledge used within these research institutions, is often appropriated from the publications generated by the research community (Dasgupta and David, 1984). Knowledge generated (output) by some scientists thus becomes knowledge appropriated (input) by other scientists. The notion of knowledge networks is clearly appropriate to the analysis of such a positive feedback system or virtuous circle. Their importance both in terms of economic growth and codified knowledge diffusion has been demonstrated (Aharonson et al., 2004), and a number of authors use social network analysis and network indicators emanating from graph theory to investigate the characteristics and degree of interrelatedness of groups of researchers (Bonaccorsi and Thoma, 2007; Cantner and Graf, 2006). Our research follows their lead and integrates social network analysis indicators of collaboration to the impact analysis.

Nanotechnology university scientists do not generally work in isolation but in increasingly large multidisciplinary research groups. Research funding is therefore rarely attributed to a single scientist, but much more commonly to a research team, especially when the amounts are rather large. The principle of collective invention, as proposed by Allen (1983), characterized by the free circulation of knowledge among socially connected agents, is very much a reality in this field. Lamoreaux and Sokoloff (1997), Dahl and Pedersen (2004), von Hippel (1987) and Schrader (1991) provide various examples of collective invention.

Theoretical simulation studies have shown that knowledge diffusion is more efficient through clustered networks (Cowan and Jonard, 2003), and relatively cliquish networks (Cowan and Jonard, 2004). The theoretical study of Cowan *et al.* (2004) shows that network structure significantly influences the long-term production of knowledge. Our paper examines these issues from an empirical point of view. In his empirical study, Newman (2001b) shows that the probability of a particular scientist acquiring new collaborators increases with the number of his

past collaborators, and that the probability of a pair of scientists collaborating increases with the number of other collaborators they have in common (the creation of a clique). A number of scholars claim that collaboration persistency through time also positively influences knowledge productivity. For instance, Cowan *et al.* (2004) claim that previous collaborations increase the probability of a successful collaboration in the future. Furthermore, Fleming *et al.* (2006) argue that an inventor's past collaboration network will strongly influence subsequent productivity.

Proposition 3: Better-positioned scientists with the co-publication network are generally more productive.

Newman (2001a) observed that for most scientists, the path linking them to other scientists goes through a relatively small number of very central individuals. These central individuals generally receive the most research funding. The question therefore is whether the position and funding of these individuals influence their knowledge production. These are the issues that we wish to address in this paper.

3 Data and methodology

3.1 Data and variables

Canada does not escape the tendencies described in the previous section in terms of increased funding, publishing and patenting of nanotechnology research. According to Hu et al. (forthcoming), Canada occupies the 5th rank in 2005-2009 with 8% of the world nanotechnology publications, an improvement from the 11th rank for the 1998-2002 period (equivalent to 2% of the world's nanotechnology publications).

The latter corresponds to a period where nanotechnology public funding was considerably fragmented (Rosei, 2008). In the subsequent years, the Natural Sciences and Engineering Research Council of Canada (NSERC) launched the five-year 5M\$ 'Nano Innovation Platform' which distributed its first research grants in 2003, while at the same time, the Canadian Institutes of Health Research (CIHR) invested roughly the same amount on regenerative medicine and nanomedicine. These initiatives obviously had an impact in raising the number of Canadian nanotechnology publications, but also of the number of patents (Dang et al., 2010), although the proportion of the world's patents that are attributed to Canada has dropped, mainly because of the rapid rise of China. It is probably too soon however to be able to really see the impact of this increased funding on patented innovation.

This research requires the integration of three data sources on funding, publishing and patenting: Elsevier's Scopus, the United States Patent and Trademark Office (USPTO) and the University Research Information System (*Système d'information sur la recherche universitaire* or *SIRU*) of the Quebec Ministry of Education, Leisure and Sports. Scopus provides the necessary data on nanotechnology scientific articles (date of publication, co-authors and their affiliations), while the USPTO provides the name and address of all inventors and patent assignees. SIRU contains the grants and contracts during the period 1985-2005 of all Quebec university scientists. The data for each scientist provides their affiliation as well as the yearly amount received from each grant or contract registered with the university¹.

¹ A number of Quebec scientists own small companies through which they perform consultative work. We have no means by which to estimate the breadth of such a practice but estimate it to be relatively small compared to the bulk of the funding received through the official channels, because of the very nature of nanotechnology research.

Using a keyword search similar to that of Porter *et al.* (2008), we have extracted all scientific articles and patents relating to nanotechnology for which at least one Canadian-affiliated scientist contributed for the period 1985-2005. This yields 4 883 patents that have been produced by 4 996 inventors and 23 250 articles authored by 24 083 scientists from which the co-publication networks will be characterized. Restricting the resulting sample to 1996-2005, because there has been a substantial change in the quality of Scopus before and after 1996, yields 17 803 articles. Merging the three databases required the careful examination of the surnames and first names (in SIRU) or initials (in Scopus and the USPTO) of scientists and their affiliations (or addresses) in all three databases. The intersection of the three databases results in the scientific production and financing of all Quebec nanotechnology academics. This exercise further reduces scientific output to 3 724 publications authored by 1 116 Quebec scientists and 566 patents involving 325 Quebec inventors. Finally, to ensure that these scientists are truly involved in nanotechnology, we select only those for which we can find at least 5 publications throughout the 21 years examined, hence eliminating the 'occasional' nanotechnology scientists. The resulting panel database contains 5 724 observations of 907 Quebec academics.

Using the social network analysis software Pajek, we then characterize the network of co-authors using co-publication links between scientists from the articles extracted. In these networks, the vertices are the co-authors or scientists and the edges between the vertices represent the coauthorship links of each article. In order to follow the evolution of collaboration over the years, we have created sub-networks using the co-publication links over periods of three and five years. An analysis of these sub-networks enabled us to describe their structural properties and to explore the collaborative behavior of nanotechnology scientists in Quebec.

An important consideration relates to the historical structure or time period of both the funding and collaboration measures. Public grants are generally awarded for a number of years. For example, Tier I Canada research chairs are generally awarded for periods of seven years (five years for Tier II chairs). Ordinary research grants vary from three to five years. Other major collaborative grants cover seven years of funding while smaller initiatives can be awarded for periods as short as one year. In the literature, the time period of collaborative networks vary from one study to the next. For instance, Schilling and Phelps (2007) use three-year windows for their firm collaborative networks. Gulati and Gargiulo (1999) prefer five-year windows and so does Stuart (2000). As a consequence, during the course of our analysis, three- and five-year subnetworks have been considered and the most robust results are reported in this paper, i.e., those for three-year sub-networks.

We seek to assess the direct impact of funding, i.e. the total amounts of grants and contracts on the number of articles published in a given year. Because of the reasons mentioned above, the average amount of funding over three and five years were included in our models and the three year averages were found to be the best suited to our data: [*AveGrant3*] and [*AveCont3*] represent respectively the average amount of grants and contracts received by an academic during the past three years. To distinguish between operating cost funds and those dedicated to infrastructure, we used the average amount of operating grants over three years [*AveGrant03*] and the average amount of infrastructure grants over three years [*AveGrant13*]².

² Negligible amounts of contracts for infrastructure acquisition are present in the database. As consequence, we do not distinguish between the infrastructure contracts and operating cost contracts.

Our measures of network attributes focus on a scientist's position within the network: betweenness centrality and individual cliquishness. Betweenness centrality of a vertex (scientist) refers to the capacity of a scientist to link two other scientists from the same three-year subnetwork through the smallest number of intermediaries [*Btwness3*]. If a greater proportion of the shortest paths between all other vertices 'goes through' a particular vertex, this vertex has a higher betweenness centrality and her role as an intermediary is more important. Cliquishness of a vertex is measured by the egocentric density of a vertex which is defined as the fraction of all pairs of the immediate neighbors of a vertex that are also directly connected to each other. The cliquishness of an individual in a three-year subnetwork [*Cliqness3*] basically measures the likelihood that two vertices that are connected to a specific third vertex are also connected to one another, hence forming a clique.

To investigate whether the relationship between patents and papers is one of reinforcement (or complementary, as Stephan et al., 2007 coined it) or of substitution (Klitkou and Gulbrandsen, 2010), we include the number of patents filed in the past three years [*nbPatent3*].

Furthermore, we add the time elapsed between the start of the nanotechnology activities of an academic in a given year to create a proxy for the nanotechnology 'career' age or experience of the researcher [*Age*]. This control variable accounts for the fact that generally, older scientists are more productive (Merton, 1973; Cole and Cole, 1973; Wray 2003 et 2004; Kyvik et Olsen, 2008). Other scholars however argue that the most extraordinary discoveries are made by researchers before their 40th birthday (Adams, 1946; Zuckerman, 1977; Stern, 1978; Gieryn, 1981).

Finally, to take into consideration university variations that would not otherwise be accounted for, we include university dummy variables [*dUniv*] for the Quebec universities: McGill University [*dUMcGill*], University of Montreal including Ecole Polytechnique of Montreal [*dUMontrealG*], Concordia University [*dUConcordia*], all the constituents of University of Quebec [*dUQG*], as well as University of Sherbrooke and Bishop University together as they are both located near Sherbrooke city [*dSherbrookeG*].

3.2 Model specification

As mentioned above, the goal of the project consists in determining whether researchers with a greater amount of grants or contracts and in better network positions generate more innovation in the form of scientific production. Our dependent variable is therefore the number of articles published in a given year [*nbArticle*]. Because the dependent variable is a count measure, we use negative binomial regressions specific to panel data to estimate the number of articles published by an individual in a given year. Appendix A provides details of the models used in this paper.

The model to be estimated can thus be expressed in reduced form as:

$$nbArticle_{it} = f \begin{pmatrix} AveGrant3_{it-1}, AveCont3_{it-1}, nbPatent3_{it-1}, \\ Btwness3_{t-2}, Cliqness3_{t-2}, Age_{it}, dUniv_i, dT_t \end{pmatrix}$$
Eq. (1)

A problem then arises because the probability of obtaining grants depends on past publications, which also depend on the grants available prior to publication. Because of the high correlation between various lags of the same variables, this problem is apparent to that of simultaneity, which is a common cause of endogeneity. A second cause of endogeneity is related to unobserved heterogeneity, which in our case would be the intrinsic quality of the researchers. Despite our efforts to collect the most complete and accurate data to account for the majority of factors that contribute to explaining the variation in the number of publications, omitted variables (such as mobility of researchers between institutions) and measurement error are inevitably present in the data at our disposal. The most important source of endogeneity however, and the one for which we will try to correct, is simultaneity. In the absence of endogeneity, the negative binomial regression model presented in Eq. (1) would be sufficient. To correct the model for endogeneity, we use the Two-Stage Residual Inclusion (2SRI) method proposed by Terza et al. (2008).

The basic idea is to estimate the endogenous variable, in our case the average amount of grants received over three years [*AveGrant3*], with an ordinary least squares regression on a number of exogenous variables or instruments. The residue of this regression is then computed and used as an explanatory variable in the model of interest. Bíró (2009) uses this particular method with a negative binomial regression for the second stage regression. Details of the methodology are provided in Appendix A.

Although we have not yet found in the literature an application of the 2SRI model using panel data, we will estimate the 2SRI model considering the overall error term as equivalent to the error term of the model utilized by Terza et al. (2008). As robustness checks we will estimate non-panel 2SRI regressions, i.e. cross-section regressions with repeated measures to account for the non-independence of the same individual through time. We will therefore show the results corresponding to the 2SRI model with and without the panel data representation, taking into account the repeated measurements of individuals through time. The STATA procedure *xtnbreg* (second stage negative binomial regression) and *xtreg* (first stage OLS regression) will be used for the panel regressions, and the procedures *nbreg* (second stage negative binomial regression)

and *reg* (first stage OLS regression) will be used with the *vce*(*cluster*) option to account for the non-independence of observations regarding the same individuals.

To correct for endogeneity, we include a number of variables that contribute to explaining the unobserved capabilities of a researcher to publish and his ability to raise public funds from grants. As grants are generally given on the basis of the past publication record of a scientist, the first instrument is the average yearly publication rate of each scientist in the three years prior to the grant [AveArticle3]. The second indicator is an ordered measure of the academic prestige linked to various research chairs. If an academic never had a chair in the period examined, the variable [codeChair] takes the value 0. If she has had an industrial chair at some point in her career, the variable takes the value 1, for NSERC or CIHR chairs, the variable takes the value 2, and finally, for Canada research chairs, the variable takes the value 3. Because we are interested in measuring the potential quality of academics, if an individual eventually reaches the top level of our chair classification, she will be attributed the top value for the entire period. Finally, the third attribute relates to the university to which an academic is affiliated. This accounts for the reputation of the establishment, the attraction power it has on different quality of individual researchers, and so on. This is accounted for by the average amount of grants raised by scientists per university in the past three years [AveGrant3U].

The regressions (first and second stages) to be estimated can therefore be expressed as:

$$AveGrant3_{it-1} = f \begin{pmatrix} AveArticle3_{it-2}, codeChair_i, AveGrant3U_{t-1}, AveCont3_{it-1}, \\ nbPatent3_{it-1}, Btwness3_{it-1}, Cliqness3_{it-1}, Age_{it}, dUniv_i, dT_t \end{pmatrix} Eq. (2)$$

$$nbArticle_{it} = f \begin{pmatrix} AveGrant3_{it-1}, AveCont3_{it-1}, nbPatent3_{it-1}, Btwness3_{it-1}, \\ Cliqness3_{it-1}, Age_{it}, dUniv_i, dT_t, residual1st \end{pmatrix} Eq. (3)$$

where Eq. (2) corresponds to Eq. (A.1) in appendix A and Eq. (2) corresponds to Eq. (A.2), and *residual1st* is the residual from Eq. (2). The model also includes year dummy variables [dT] to account for any yearly differences that would not be accounted for by the explanatory variables.

4 Descriptive statistics

Before turning to the regression results, let us first briefly describe some of the explanatory variables (which are further described in appendix B at the end of the paper). Fig. 1 presents the average number of articles per academic, our dependent variable, for each of the universities examined. McGill University generally dominates, followed by Laval University and the University of Montreal in second and third places. Concordia University rises sharply towards the end of the sample, despite a declining average amount of public funds received (Fig. 2).

Insert Fig. 1 approximately here

Insert Fig. 2 approximately here

Examining the average amount of research funds granted to individual researchers (Fig. 2) shows that until the turn of the century, the differences between the amounts received by scientists from different universities is not as pronounced as in the latter part of the sample. The peaks obtained by University of Montreal in 2000 and 2003, as well as by Laval University in 2002 and by McGill University in 2004 are mainly due to large infrastructure grants obtained by their scientists.

With the recruitment of new professors and the use of the Canadian Foundation for Innovation (CFI), these universities have built a considerable research infrastructure. The recent increase in scientific production (though not apparent from the graphs which show averages per scientist) is

probably attributable to this investment. Scientists from Concordia, Sherbrooke and Quebec universities receive less funds per year than their colleagues of McGill, Montreal and Laval universities. Fig. 3 indicates that researchers from the University of Montreal have on average received almost double the industrial contracts of McGill University. This would tend to suggest more applied work for/with industry. In that game, Laval University wins hands down with the highest proportion of contracts among all Quebec universities.

Turning now to the network measures, we find that both betweenness centrality and cliquishness have decreased slightly over the years (see Fig. 4) but that the former has stabilized since the turn of the century. As more scientists from a number of different disciplines turn their attention to nanotechnology, it is to be expected that the betweenness centrality of individuals will decrease over time. This phenomenon is also observed for nanotechnology patents over the same period by Beaudry and Schiffauerova (2011). The authors link this diminution over time to the increased fragmentation of the field due to its development in a vast array of domains of application.

Insert Fig. 4 approximately here

Because we used the variable [*Age*] as is a proxy for the experience gained in nanotechnology by the researcher over time, we examine the interaction of this variable with the number of articles and the number of patents (Fig. 5). As mentioned by Lee and Bozeman (2005) and Lehman (1953), research productivity grows with the age of the scientist and peaks at a certain period. The latter vary with the discipline and whether the research is fundamental or more technology-oriented. In our case, the age of the scientific production continues to increase with age, while

patenting tends to follow an inverted U-shaped curve. Both curves imply non-linear effects, which will be taken into consideration in the regressions that follow.

Insert Fig. 5 approximately here

5 Results

The regressions estimate the factors that influence researcher productivity measured by the number of articles (the results are shown in Table 1). Our analysis has considered a lag structure of one-, two- and three-year lags for most variables of the model, similar to what Schilling and Phelps (2007) have used in their model. We have tested all combinations of lags and the most robust results combine one- and two-year lags (one-year lag was found to be most appropriate for public grants and contracts while two-year lags were more robust for the network variables). We also included interaction effects between variables, and a quadratic term for others to account for non-linear effects.

Table 1 presents the results of 10 regression models, using a number of variables to represent grants, contracts and network position indicators. The first two models do not consider the endogeneity; they estimate Eq. (1). Model (1) uses a panel data negative binomial regression model while model (2) estimates a cross-section negative binomial model (i.e. without the panel data representation) using a clustering method to include repeated observations of the same scientist over the years. Models (3) to (8) are the second stage regressions, Eq. (3), of the 2SRI method with a panel data negative binomial regression. We tested different sets of variables over a hierarchical progression for the inclusion of the variables. The last two models, (9) and (10), estimate the second stage regression negative binomial model (without

the panel data structure) using the same clustering method. Table C.1 in the appendix presents the related results for the first stage regressions, Eq. (2), of the 2SRI models.

Insert Table 1 approximately here

Insert Table 2 approximately here

Let us now take each variable in turn and analyze its influence on scientific production. First, the influence of the average amount of grants over three years [AveGrant3] on the number of articles published follows a quadratic convex function (J-shaped), which implies that the productivity of the researcher increases exponentially (the linear effect is not significantly different from 0). This indicates that larger grants, or more funding, increase the number of published articles, a result in accordance with Arora et al. (1998) and Zucker et al. (1997). The results are similar if we distinguish operating cost grants [AveGrantO3] and infrastructure grants [AveGrantI3] in Table 2. This last variable only appears in the first stage regression as it is not correlated with the number of articles, nor was it significant in the second stage regression (this contrasts with Payne and Siow, 2003 who find a direct effect of infrastructure grants on scientific output). The average amount of infrastructure grants is therefore added as an instrument in the first stage regressions and contributes to explaining the average amount of operating cost grants received by an academic. Its influence on the amount of operating cost grants vary linearly (see Table C.2 in the appendix). As with the average amount of public funds, the average amount of operating cost grants raised by a scientist has a positive effect on the productivity of the researcher. Operating cost grants are destined for the direct resources working on the research (students, laboratory assistant, research assistant, equipment maintenance, etc.), which are important for the discovery of any knowledge worth publishing. Through the different models, we can say that the simple

(linear) effect of the total amount is positive and significant, but that its effect is closer to exponential as shown by the coefficient of the quadratic effect introduced.

Turning now to our instruments to correct for potential endogeneity of the average amount of grants received, in Eq. (2), we generally find them to have a positive and significant effect. These results are presented in Table C.1 in the appendix (Table C.2 presents the corresponding results for the distinction between operating cost and infrastructure grants). A greater yearly average of publications in the past three years has the expected positive influence on the likelihood of raising more public research funds, and so does the chair 'prestige' status. These findings obviously follow the general intuition and more specifically support those of Arora et al. (1998). The average amount of research funds raised by other university colleagues of the same institution also contributes to explaining our endogenous variable. In these regressions, we also find that researchers that raise more funds from private contracts also raise more funds from public sources.

Generally, private funds are destined for specific projects in relation to an industrial problem. This work is considered 'protected' for the industry as it may contain confidential content. That's why companies set restrictions for the research results that would most of the time prevent a scientist from publishing articles related to the output of his industry collaboration. The output of privately funded research tends to appear more as patents, because it guarantees ownership of the intellectual property to the investors. The total amount of contracts does not however appear to have a significant effect (neither negative, nor positive) on the number of publications, only in one of the cross-section (regression 9) is it negative and significant. This finding contradicts the literature that suggests a reinforcing effect of private funds on research productivity (Gulbrandsen and Smeby, 2005; Van Looy et al., 2004), and the literature suggesting a

substitution effect (anticipated for instance by Welsh et al., 2008 and Blumenthal et al., 1996). Our findings may be explained by the fact that nanotechnology is still relatively far from the market. Because the field is quite young, private contracts may result in scientific articles, but they may also yield patents in the fields closer to technology applications (the evolution from microelectronics to nanoelectronics for example).

For this reason, we have tested whether there was a moderating effect between patents and private contracts, but to no avail, the resulting interactive variable was never significant. Although we find no evidence of impact on scientific production, the very nature of nanotechnology calls for this closeness between industry and universities. Canada has invested vast amounts on infrastructure and laboratory equipment, mostly from public sources. These expensive resources (both in infrastructure and in the skilled labor necessary for its operation) would be difficult to finance privately and hence contribute to attracting industries (Robinson et al., 2007 for instance examine cluster formation enabled by these infrastructures or platforms). The interaction between publicly funded research infrastructure and private contracts is therefore important to examine.

Both variables [*AveGrant13* and *AveCont3*] in Table C.1 and Table C.2 have a positive and significant influence on the average amount of public grants raised by a scientist. In fact, having the laboratory machinery and the equipment necessary for high technology research is a strong argument to attract the grants to operate them. Similarly, the average amount of contract seems to attract more operating cost funds to a researcher. The positive and significant effect of the contract variable suggests a Matthew effect (money attracts more money) that is very much verified in our data set (Merton, 1968). Arora et al. (1998), suggest that funding is more likely to be directed towards scientists with an established tracked record of successes, and we would add

a tracked record that was the result of more funding. The interaction between the two variables, however, has a negative and significant effect, implying that the infrastructure used in conjunction with private firms reduces the need to obtain public funding. In regards to the use of infrastructure, public and private funds therefore appear to be substitutes.

Industrial interests or potential commercialization does nevertheless have an influence as the number of patents has an inverted-U relationship with the number of articles. The maximum of the resulting curve corresponds roughly to 30 patents over three years. As a consequence, before that point, the first patents tend to reinforce an individual's scientific production and strengthen the reputation of the researcher. Beyond that point, however, contributing to more patents implies a declining number of articles published, *ceteris paribus*. Hence patents are detrimental to scientific productivity only beyond a relatively large amount of patent applications filed over a period of three years. Once a rather large number of patent applications have been made, scientists tend to prioritize applied work and focus on patenting; it clearly becomes a career choice, which may then be more profitable.

Regarding the influence of collaboration, we find that betweenness centrality [*Btwness3*] has a positive and significant effect on the number of articles. This confirms our expectation that a researcher with a more central position in the co-publication network, i.e. a more influential intermediary, generally publishes more articles. In addition, past individual cliquishness [*cliqness3*] tends to have a negative effect on scientific productivity. Once we account for nonlinearities of this variable, we find what appears like the left-hand branch of an inverted-U relationship with the number of published articles, i.e. scientific productivity exhibits diminishing returns to increased cliquishness. A higher cliquishness value implies that the collaborators of an academic are more likely to be collaborators as well, i.e. to form a clique. Our

results would tend to suggest that if a researcher maintains some degree of segregation between the groups with whom he collaborates, which would yield a lower cliquishness value, his probability of publishing is greater if his 'clique' is slightly more integrated. Further along the curve of increased cliquishness, however, the returns of a greater integration of the clique are much smaller. In some fields, large multidisciplinary teams are necessary and, as a consequence of the integration of knowledge from various sources, publication is slightly slower as various disciplines adjust to each other. When a field is in its infancy, as nanotechnology clearly is, the skills and knowledge necessary for research probably go beyond the restricted circle of a scientist's direct collaborators, especially when the resulting technology has a vast array of potential applications. This would explain, to some extent, why working in increasingly cliquish nanotechnology networks do not tend to augment the productivity of each scientist as much as when there is a rather fragmented clique around a scientist. This would tend to support an extension to scientific productivity of Cowan and Jonard's (2004) argument that knowledge diffusion is more efficient in 'relatively' cliquish networks, i.e. not too cliquish but cliquish enough.

Examining closely regressions (3), (4) and (5) shows that betweenness centrality interacts with other variables in our models and moderates their influence on the dependent variable, particularly patenting and cliquishness. To counter this intrinsic relation of betweenness centrality with these variables, we have introduced interactive variables with cliquishness, the average amount of contracts and the number of patents³. Interacting betweenness centrality with cliquishness the diminishing returns impact of the latter. In contrast, interacting

³ Other interactions have been tested, but were non significant. Only those presented in Table 1 are significant and are worth including in the models.

betweenness centrality with the total number of patents over three years contributes to dampening the inverted-U shaped effect of patenting on scientific productivity.

Our last explanatory variable, the nanotechnology 'career' age of a scientist, generally has a positive effect on scientific productivity, in line with Fig. 5. When we add a quadratic term, the curvature of the relationship appears. Around 16 years after starting nanotechnology work, scientific production starts to decline. It would therefore appear that it is the mid-career scientists that have the highest scientific production, a result in line with that of Cole (1979), Wray (2003 and 2004) as well as Kyvik and Olsen (2008).

A recent debate (Wells, 2009) on university research funding in Canada saw the 'big 5' universities (University of British Columbia, University of Alberta, University of Toronto, University of Montreal and McGill University) argue that they should receive all (or the bulk of) the research funding, concentrate only on graduate training and leave undergraduate teaching to the other Canadian universities. The argument put forward in Wells' (2009) article by the five top university presidents is that they should be allowed to "pursue world-class scientific research and train the most capable graduate students".

Although this proposal raised a wall of protestation in Canada, the question as to where does research funding yields the highest scientific productivity deserves an answer. In this article, we focus on the Quebec universities and compare the publication results obtained by two of the 'big 5' (University of Montreal and McGill University) as a result of the research funding received. From the analysis of the university dummy variables included in the regressions, we cannot say that our results support their claim, certainly not in terms of scientific production. The omitted dummy variable is for McGill University, to which the results should be compared. All

university dummy variables are significantly different from McGill University with the exception of Concordia University, which is sometimes weakly significant. In the field of nanotechnology, the University of Montreal does not appear to deserve its 'title': Being located at the University of Montreal has no significantly different effect than being located at Laval University (in Quebec city). We can say that there is no evidence so far that researchers from the two 'best' universities in Quebec publish 'more' than their colleagues in other universities in the province. The 'publish or perish game' is similar for all universities and researchers conform to it.

6 Discussion and conclusion

At the beginning of this paper, we set out to investigate whether the funding and organization (read collaboration or co-authorship) of research has an impact on the output from this research. To that effect, we suggested three propositions, two on funding and one on the individual network position of an academic.

To the first part of the question, on the funding, the answer is overwhelmingly 'yes there is an impact', but there are subtleties. While more public research funds undoubtedly lead to more scientific articles, hence supporting proposition 1 and the work of other scholars (for instance Arora et al., 1998, Zucker et al. 2007), the relationship between private funds and scientific production is inexistent (in contrast with Van Looy et al., 2004), hence rejecting proposition 2. Contracts, which are generally associated with more applied research, do not have the commonly found reinforcing impact on the scientific production in nanotechnology. Public policy should not therefore consider industrial contracting as a threat to the creation of a collective knowledge base, not at the current level of industry involvement in any case.

Since the amount of private research funds at the disposal of researchers does not yield a convincing story, the number of patents to which a researcher has contributed may play a more fundamental role. We indeed find that it is not so much the fact that academics collaborate with firms that affects their publishing record, it is too much patenting. And when we mean too much, we imply more than 20 to 40 patents over a period of three years, which clearly is a career choice. Below that extraordinary contribution to technology, we clearly find a reinforcement effect between patenting and publishing, hence supporting the findings of a number of scholars (Breschi et al., 2005; Calderini and Franzoni, 2004; Calderini et al., 2007; Meyer, 2006; Van Looy et al., 2006). The implication in terms of public policy would be to encourage, or at least not discourage, scientists to bring their research closer to the market as the protection of intellectual property does have a positive impact on their scientific production when performed within reasonable bounds, bearing in mind that too much of a good thing can be detrimental to scientific production.

In a field such as nanotechnology, where innovation increasingly requires multidisciplinary teams, innovating alone is no longer possible. The most 'important' scientists should thus occupy the most central positions within the co-authorship networks. Betweenness centrality acts as a proxy for a researcher' importance as an intermediary in our impact regressions. We show that better intermediaries have a better scientific performance. In addition, the structure of research teams, measured by the cliquishness, or clustering coefficient, also has an influence on scientific output. The maximum cliquishness, however, does not bring as much productivity benefits as we would expect, the relationship between cliquishness and scientific production exhibiting diminishing returns. In other words, one's efforts in integrating team members into a more cliquish network may not be fully rewarded. Some level of fragmentation in the immediate

surrounding network of a scientist is therefore acceptable. To our knowledge, this is quite a novel result which raises some interesting questions regarding the optimal size and diversity of teams, compared with the amount of effort, expense and coordination that is required for the management of such teams. For a number of years now, Canada has promoted networking and multidiciplinarity via its granting councils, it is now time to pause and to measure the impacts of such policies in terms of scientific output and technology as excessive networking may not be as beneficial as one would have thought.

There are a number of limitations to our research. First, we have examined a rather narrow field, nanotechnology. In order to fully support the claims of two of the 'big 5' universities in Quebec, for instance, one would have to assess all the disciplines in which both universities specialize. A second limitation concerns the mobility of scientists. Once a scientist moves out of the province, he disappears from our funding radar even though he may remain active. A third limitation of our study is the data. Although we have tried to extract as much of the articles of the researchers identified in the nanotechnology fields, it is possible that some escaped our net. Scopus does not include all scientific journals and as more journals are added every year, the coverage changes constantly, adding a time bias to the analysis. A fourth limitation stems from the fact that our model cannot measure the impact of graduate students on the production of a research group, because they are not academic professors employed at a university, they appear to have no funding. Finally, we have not yet been able to explicitly take into account research teams apart from those measured by co-publications.

This last limitation exemplifies an important knowledge gap on research teams. and raises a number of interesting questions regarding the organization of research between localized and delocalized teams, and how public funding contributes to the constitution of these teams,

especially in light of the new 'networking' policies. Is more knowledge generated as a consequence of these policies? Similarly, we need a more comprehensive examination of whether private funding and the interaction with industry modify the composition of these research teams. Thursby and Thursby (2011), for instance, have made an interesting contribution regarding various interactions with industry and their impact. In other words, we need to evaluate whether the changes imposed on researchers by grant awarding bodies are a step in the right direction.

Acknowledgements

Beaudry acknowledges financial support from the Social Science and Humanities Research Council of Canada (grant # 421-2007-1021) and the Canadian Institutes of Health Research (grant # KRS-94306). Earlier versions of this work (on nanotechnology and biotechnology) were presented at DRUID 2010 conference in London, the 2010 Triple Helix conference in Glasgow, the 2011 DIME workshop in Turin and the Atlanta conference on Science and Innovation Policy in 2011. We are grateful to Carl St-Pierre for his advice on statistical analysis, to Ahmad Barirani for the data extraction program used in the data collection, to Maxime Clerk-Lamalice for the construction of the data consolidation program and for the exploratory statistical work, and to Ramine Kananian and Ricard-Olivier Moreau for their contribution to the long and painful database matching process. We would like to thank the following individuals who have commented on earlier drafts of this research: Joachim Henkel, Martin Kenney, Katrina Larsen, Cornelia Meissner, Markus Perkmann, two anonymous referees as well as the participants of the conferences listed above. None of these, however, are responsible for any remaining errors.

References

Adams, C.W., 1946. The age at which scientists do their best work. Isis 36, 166–169.

- Adams, J. D., Black, G. C., Clemmons, J. R., Stephan, P. E., 2005. Scientific teams and institutional collaborations: Evidence from U.S. universities, 1981-1999. Research Policy 34, 259-285.
- Agrawal, A., Henderson, R., 2002. Putting Patents in Context: Exploring Knowledge Transfer from MIT. Management Science 48, 44-60.

- Aharonson, B. S., Baum, J. A. C., Feldman, M. P., 2004. Industrial clustering and the returns to inventive activity: Canadian biotechnology firms, 1991-2000. DRUID Working paper No. 04-03.
- Allen, R. C., 1983. Collective invention. Journal of Economic Behavior and Organization 4, 1-24.
- Arora, A., David, P. A., Gambardella, A., 1998. Reputation and Competence in Publicly Funded Science: Estimating the Effects on Research Group Productivity. Annales d'Économie et de Statistique 49/50, 163-198.
- Audretsch, D. B., Stephan, P. E., 1996. Company-scientist location links: the case of biotechnology. American Economic Review 86, 641-652.
- Balconi, M., Laboranti, A., 2006. University–industry interactions in applied research: The case of microelectronics. Research Policy 35, 1616-1630.
- Barjak, F., Robinson, S., 2007. International collaboration, mobility and team diversity in the life sciences: Impact on research performance. Social Geography Discussion 3, 121-157.
- Beaudry, C., Schiffauerova, A., 2011. Impacts of collaboration and network indicators on patent quality: The case of Canadian nanotechnology innovation. European Management Journal 29, 362-376.
- Bíró, A., 2009. Health care utilization of older people in Europe Does financing structure matter?. Working paper, Central European University.
- Blume-Kohout, M., Kumar, K., Sood, N., 2009. Federal life sciences funding and university R&D. NBER Working Papers No. 15146.

- Blumenthal, D., Campbell, E. G., Causino, N., Louis, K. S., 1996. Participation of Life-Science Faculty in Research Relationships with Industry. New England Journal of Medicine 335(23), 1734-1739.
- Bonaccorsi, A., Thoma, G., 2007. Institutional complementarity and inventive performance in nano science and technology. Research Policy 36, 813-831.
- Breschi, S., Lissoni, F., Montobbio, F., 2005. From publishing to patenting: Do productive scientists turn into academic inventors?. Revue d'Economie Industrielle 110, 75-102.
- Calderini, M., Franzoni, C., 2004. Is academic patenting detrimental to high quality research? An empirical analysis of the relationship between scientific careers and patent applications.
 CESPRI Working Paper No. 162. Università Commerciale Luigi Bocconi, CESPRI, Milano.
- Calderini, M., Franzoni, C., Vezzulli, A., 2007. If star scientists do not patent: The effect of productivity, basicness and impact on the decision to patent in the academic world. Research Policy 36, 303-319.
- Cantner, U., Graf, H., 2006. The network of innovators in Jena: An application of social network analysis. Research Policy 35, 463-480.
- Cohen, W. M., Nelson, R. R., Walsh, J. P., 2002. Links and Impacts: The Influence of Public Research on Industrial R&D. Management Science 48, 1-23.
- Cole, J. R., Cole, S., 1973. Social Stratification in Science, The University of Chicago Press, Chicago, IL.
- Cole, S., 1979. Age and scientific performance. American Journal of Sociology 84, 958-977.
- Cowan, R., Jonard, N., 2003. The dynamics of collective invention. Journal of Economic Behaviour and Organization 52, 513-532.

- Cowan, R., Jonard, N., 2004. Network structure and the diffusion of knowledge. Journal of Economic Dynamics and Control 28, 1557-1575.
- Cowan, R., Jonard, N., Ozman, M., 2004. Knowledge dynamics in a network industry. Technological Forecasting and Social Change 71, 469-484
- Dahl, M. S., Pedersen, C. O. R., 2004. Knowledge flows through informal contacts in industrial clusters: myth or reality?. Research Policy 33, 1673-1686.
- Daniel, H. Z., Hempel, D. J., Srinivasan, N., 2003. Project selection: A process analysis. Industrial Marketing Management 32, 39-54.
- Dang, Y., Zhang, Y., Fan, L., Chen, H., Roco, M. C., 2010. Trends in worldwide nanotechnology patent applications: 1991 to 2008. Journal of nanoparticle research 12, 687-706.
- Dasgupta, M. R., David, P. A., 1994. Toward a new economics of science. Research Policy 23, 487-521.
- Fleming, L., King, III, C., Juda, A. I., 2007. Small Worlds and Regional Innovation, Organization Science 18, 938-954.
- Gieryn, T.F., 1981. The aging of a science and its exploitation of innovation: Lessons from X– ray and radio astronomy. Scientometrics 3, 325–334.
- Gibbons, M., Limoges, C., Nowotny, H., Schwartzman, S., Scott, P., Trow, M. 1994. The New Production of Knowledge. Sage.
- Goldfarb, B., 2008. The effect of government contracting on academic research: Does the source of funding affect scientific output? Research Policy 37, 41-58.

- Gulati, R., Gargiulo, M., 1999. Where Do Interorganizational Networks Come From?. American Journal of Sociology 104, 1398-1438.
- Gulbrandsen, M., Smeby, J., 2005. Industry funding and university professors' research performance. Research Policy 34, 932–950.
- Hagedoorn, J., Link, A. N., Vonortas, N. S., 2000. Research partnerships. Research Policy 29, 567-586.
- Hall, B. H., Link, A. N., Scott, J. T., 2003. Universities as research partners. Review of Economics and Statistics 85, 485-491.
- Hart, D. M., 2001. Antitrust and technological innovation in the US: ideas, institutions, decisions, and impacts, 1890-2000. Research Policy 30, 923-938.
- Hausman, J., Hall, B. H., Griliches, Z., 1984. Econometric models for count data with an application to the patents-R & D relationship. Econometrica 52, 909-938.
- Hu, G., Carley, S., Tang, L. (forthcoming). Visualizing nanotechnology research in Canada: evidence from publication activities, 1990–2009, *Journal of Technology Transfer*, published online 2011-11-09.
- Jacob, B., Lefgren, L., 2007. The impact of research grant funding on scientific productivity. NBER Working Papers No. 13519.
- Katz, J.S., Hicks, D., 1997. How much is a collaboration worth? A calibrated bibliometric model. Scientometrics 40, 541–554.
- Klitkou, A., Gulbrandsen, M., 2010. The relationship between academic patenting and scientific publishing in Norway. Scientometrics 82, 93-108.

- Kyvik, S., Olsen, T. B., 2008. Does the aging of tenured academic staff affect the research performance of universities?. Scientometrics 76, 439–455.
- Lamoreaux, N. R., Sokoloff, K. I., 1997. Location and technological change in the American glass industry during the late nineteenth and early twentieth century. NBER Working Papers No. 5938.
- Landry, R., Traore, N., Godin, B., 1996. An econometric analysis of the effect of collaboration on academic research productivity. Higher Education 32, 283–301.
- Larsen, M. T., 2011. The implications of academic enterprise for public science: An overview of the empirical evidence. Research Policy 40, 6-19.
- Lee S., Bozeman, B., 2005. The impact of research collaboration on scientific productivity. Social Studies of Science 35, 673-702.
- Lehman, H. C., 1953. Age and Achievement. Princeton University Press, Princeton, PA.
- Link, A. N., Scott, J. T., 2004. The Role of Public Research Institutions in a National Innovation System: An Economic Perspective, World Bank White Paper.
- Lipsey, R., Carlaw, K., Bekar, C., 2005. Economic Transformations General Purpose Technologies and Long-Term Economic Growth. Oxford University Press, Oxford, UK.
- Martín-Sempere, M. J., Rey-Rocha, J., Garzon-Garcia, B., 2002. The effect of team consolidation on research collaboration and performance of scientists. Case study of Spanish University researchers in Geology. Scientometrics 55, 377–394.
- Merton, R. K., 1968. The Matthew effects in science, Science (January 5), 56-63.

- Merton, R. K., 1973. The Sociology of Science: Theoretical and Empirical Investigations. Chicago and London: Chicago University Press.
- Meyer, M., 2006. Knowledge integrators or weak links? An exploratory comparison of patenting researchers with their non-inventing peers in nano-science and technology. Scientometrics 68, 545-560.
- Mogoutov, A., Kahane, B., 2007. Data search strategy for science and technology emergence: A scalable and evolutionary query for nanotechnology tracking. Research Policy 36, 893-903.
- Newman, M. E. J., 2001a. Scientific collaboration networks. II. Shortest paths, weighted networks, and centrality. Physical Review 64, 016132-1-7.
- Newman, M. E. J., 2001b. Clustering and preferential attachment in growing networks. Physical Review 64, 025102.
- Owen-Smith, J., Powell, W. W., Steven, V., 2001. Careers and contradictions: faculty responses to the transformation of knowledge and its uses in the life sciences, in: Valllas, S. (Ed.), Research in the Sociology of Work 10, pp. 109-140.
- Palmberg, C., Dernis, H., Miguet, C., 2009. Nanotechnology: An overview based on indicators and statistics. OECD Science, Technology and Industry Working Papers, 2009/7, OECD Publishing.
- Payne, A. A., Siow, A., 2003. Does federal research funding increase university research output?. Advances in Economic Analysis & Policy 3, 1-22.
- Porter, A. L., Youtie, J., Shapira, P., Schoeneck, D. J., 2008. Refining search terms for nanotechnology. Journal of Nanoparticle Research 10, 715-728.

- Robinson, D. K. R., Rip, A., Mangematin, V., 2007. Technological agglomeration and the emergence of clusters and networks in nanotechnology. Research Policy 36, 871-879.
- Rosei, F., 2008. Editorial of the Special Issue on Nanotechnology in Canada. International Journal of Nanotechnology 5, 897-899.
- Schilling, M.A., Phelps, C.C., 2007. Interfirm collaboration networks: The impact of large-scale network structure on firm innovation. Management Science 53, 1113-1126.
- Schrader, S., 1991. Informal technology transfer between firms: Cooperation through information trading. Research Policy 20, 153-170.
- Stephan, P. E., Gurmu, S., Sumell, A. J., Black, G., 2007. Who's patenting in the university? Evidence from the survey of doctorate recipients. Economics of Innovation and New Technology 16, 71–99.
- Stern, S., 1978. Age and achievement in mathematics: A case–study in the sociology of science. Social Studies of Science 8, 127–140.
- Stuart, T. E., 2000. Interorganizational alliances and the performance of firms: A study of growth and innovation rates in a high-technology industry. Strategic Management Journal 21, 791-811.
- Terza, J. V., Basu, A., Rathouz P. J., 2008. Two-stage residual inclusion estimation: Addressing endogeneity in health econometric modelling. Journal of Health Economics 27, 531-543.
- Thursby, J., Thursby M., 2011. University-industry linkages in nanotechnology and biotechnology: evidence on collaborative patterns for new methods of inventing. The Journal of Technology Transfer 36, 605-623.

- Van Looy, B., Callaert, J., Debackere, K., 2006. Publication and patent behavior of academic researchers: Conflicting, reinforcing or merely co-existing?. Research Policy 35, 596-608.
- Van Looy, B., Ramga, M., Callaert, J., Debackere, K., Zimmermann, E., 2004. Combining entrepreneurial and scientific performance in academia: towards a compounded and reciprocal Matthew-effect?. Research Policy 33, 425-441.
- von Hippel, E., 1987. Cooperation between rivals: Informal know-how trading. Research Policy 16, 291-302.
- Wells, P., 2009. Our universities can be smarter, *Macleans*, July 28. Available at: http://www2.macleans.ca/2009/07/28/our-universities-can-be-smarter/
- Welsh, R., Glenna, L., Lacy, W., Biscotti, D., 2008. Close enough but not too far: Assessing the effects of university-industry research relationships and the rise of academic capitalism.Research Policy 37, 1854-1864.
- Wray, K.B., 2003. Is science really a young man's game?. Social Studies of Science 33, 137– 149.
- Wray, K. B., 2004. An examination of the contributions of young scientists in new fields. Scientometrics 61, 117–128.
- Youtie, Y., Iacopetta, M., Graham, S., 2008. Assessing the nature of nanotechnology can we uncover an emerging general purpose technology?. Journal of Technology Transfer 33, 315-329.
- Zucker, L. G., Darby, M. R., Armstrong, J., 1998. Geographically localized knowledge: spillovers or markets?. Economic Inquiry 36, 65-86.

- Zucker, L. G., Darby, M. R., Armstrong, J., 2002. Commercializing knowledge: university science, knowledge capture, and firm performance in biotechnology. Management Science 48, 138-153.
- Zucker, L. G., Darby, M. R., Furner, J., Liu, R. C., Ma, H., 2007. Minerva unbound: Knowledge stocks, knowledge flows and new knowledge production. Research Policy 36, 850-863.
- Zuckerman, H.A., 1977. Scientific Elite: Nobel Laureates in the United States. Free Press, New York, NY.

Appendix A – Model specifications

Our dependent variable [*nbArticle*] being a count measure for each individual scientist over the years, a Poisson regression is generally appropriate for this purpose (Hausman *et al.*, 1984)

$$\Pr(Y_{it} = y_{it}) = \exp \lambda(x_{it}) \cdot \left[\frac{\lambda(x_{it})}{y_{it}!}\right].$$

The model expresses the probability of the number of occurrence of the observed value y_{it} , i.e. the number of articles, depending on the parameter λ that is a function of the explanatory variables x_{it} . This model implies the strong assumption that the variance of the number of occurrences is equal to the expected number of occurrences $Var[Y_{it}] = E[Y_{it}]$. This assumption further implies that there is no over-dispersion (when the variance exceeds the mean) in the sample. This over-dispersion causes for the standard errors to be underestimated, and hence for the significance of the coefficients to be overestimated. The negative binomial model is a solution to this over-dispersion, in which the parameter λ is expressed as $\lambda = \exp(\pi x_{it})\varepsilon$, where the error ε follows a Gamma distribution. This leads to write the variance as $Var[Y_{it}] = E[Y_{it}](1 + \alpha E[Y_{it}])$, where α is the parameter of the Gamma distribution.

In order to account for endogeneity within the negative binomial regression model, we use the Two-Stage Residual Inclusion (2SRI) method proposed by Terza et al. (2008). The main assumption of the model is that the conditional mean of the dependent variable can be written as (from Terza et al., 2008) $E[y_i|x_{in}, x_{ix}, x_{iu}] = M(x_{in}\beta_n + x_{ix}\beta_x + x_{iu}\beta_u)$, where $M(\cdot)$ is a non linear function, x_{in} is a vector of endogenous variables (in our case, there is only one), x_{ix} is a vector of exogenous variables, and x_{iu} is a vector of unobservable latent variables (omitted variables)

that has an effect on the dependent variable and is correlated with the endogenous variable. The corresponding regression model can be written as $y_i = M(x_{in}\beta_n + x_{ix}\beta_x + x_{iu}\beta_u) + u_i$, where u_i is the error term.

The first regression (called auxiliary regression) therefore consists in modelling the relationship between the endogenous variable and the unobserved latent variables using a set of instrumental variables (IV), w_i , that can be written as follows (in this case, we consider a linear relationship):

$$x_{in} = w_i \alpha + x_{iu}$$
 Eq. (A.1)

where $w_i = [x_{in} w^+]$ and w^+ is the vector of instrumental variables and α is the vector of parameters. These variable have to meet the three following conditions: first, the instrumental variables must not be correlated with the unobserved variables $E[x_u|w_i] = 0$; second, the instrumental variables must be sufficiently correlated with the endogenous variable, which means that the instrumental variables would be sufficient to estimate the endogenous variable; and third, they are not correlated with the main variable of interest, nor with the error term.

The first stage consists in estimating the auxiliary regression by ordinary least squares (OLS) and calculating the predicted values, $\hat{x}_{in} = w_i \hat{\alpha}$, where $\hat{\alpha}$ is the estimate of α for the first equation (auxiliary regression). Subsequently, the residuals of the unobserved variables are calculated by $x_{iu} = x_{in} - \hat{x}_{in}$.

The second stage then consists in estimating the negative binomial regression model by introducing the residues as an explanatory variable. In the case of panel data, the error term contains both u_i and a time related error component e_{it} .

$$y_{it} = M\left(x_{int}\beta_n + x_{ixt}\beta_x + x_{iut}\beta_u\right) + u_i + e_{it}$$

Eq. (A.2)

Appendix B – Descriptive statistics

Insert Table B.1 approximately here

Insert Table B.2 approximately here

Appendix C – Additional regression results

Insert Table C.1 approximately here

Insert Table C.2 approximately here



Fig. 1 – Average number of articles per academic published per year



Fig. 2 – Average amount of public funds received per academic (in constant Canadian dollars of 2002) per year



Fig. 3 – Average amount of public (operating costs grants and infrastructure grants) and private (contracts) funds received (in constant Canadian dollars of 2002) per scientist for the period 1996-2005



Fig. 4 – Average betweenness centrality (*Btwness3* x 10⁴) and cliquishness (*Cliqness3* x 10⁴) per year



Fig. 5 – Index (relative to the maximum value over the years) of the number of articles and patents by the nanotechnology 'career' age of scientists

Table 1 –	Negative binomia	l regression results -	- number of	articles per	academic per year
-----------	------------------	------------------------	-------------	--------------	-------------------

Model Panel X-section Panel Two-Stage Residual Inclusion (2SRI)								X-section 2SRI		
Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Average amount of gran	ts over the pas	t 3 years								
$\ln(AveGrant\mathcal{Z}_{t-1})$	-0.0628 *** (0.0141)	-0.0877 *** (0.0189)	0.0458 *** (0.0183)	0.0498 *** (0.0180)	0.0438 *** (0.0180)	-0.0201 (0.0252)	-0.0322 (0.0251)	-0.0302 (0.0251)	0.2257 *** (0.0339)	0.1331 *** (0.0386)
$\left[\ln(AveGrant\mathcal{Z}_{t-1})\right]^2$	0.0052 *** (0.0011)	0.0081 *** (0.0016)				0.0049 *** (0.0011)	0.0049 *** (0.0011)	0.0048 *** (0.0011)		0.0058 *** (0.0015)
Average amount of contr	acts over the	past 3 years								
$\ln(AveCont\mathcal{Z}_{t-1})$	-0.0100 (0.0136)	-0.0047 (0.0188)	0.0038 (0.0035)	0.0046 (0.0035)	0.0053 (0.0036)	-0.0112 (0.0137)	-0.0110 (0.0136)	-0.0135 (0.0137)	-0.0137 *** (0.0057)	-0.0251 (0.0182)
$[\ln(AveCont\mathcal{Z}_{t-1})]^2$	0.0015 (0.0013)	0.0010 (0.0018)				0.0013 (0.0013)	0.0016 (0.0013)	0.0016 (0.0013)		0.0008 (0.0017)
Number of patent applic	ations of the p	ast 3 years								
$nbPatent3_{t-1}$	0.0777 *** (0.0136)	0.0902 *** (0.0210)	0.0177 *** (0.0044)	-0.0034 (0.0049)	0.0500 *** (0.0084)	0.0669 *** (0.0110)	0.0579 *** (0.0116)	0.0800 *** (0.0136)	0.0518 *** (0.0132)	0.0960 *** (0.0194)
$[nbPatent3_{t-1}]^2$	-0.0012 ** (0.0005)	-0.0020 ** (0.0008)				-0.0015 *** (0.0002)	-0.0006 (0.0005)	-0.0013 ** (0.0005)		-0.0023 *** (0.0007)
3-year network betweenn	ess centrality									
$Btwness3_{t-2} \ge 10^{-1}$	0.0426 *** (0.0067)	0.1238 *** (0.01474)		0.0101 *** (0.0011)	0.0395 *** (0.0067)	0.0111 *** (0.0011)	0.0412 *** (0.0068)	0.0414 *** (0.0067)	0.08/1 *** (0.0180)	0.0993 *** (0.0147)
3-year network cliquishn	ess									
$\ln(Cliqness \mathcal{Z}_{t-2} \times 10^4)$	0.0687 ** (0.0309)	0.2075 *** (0.0450)	-0.0045 * (0.0025)	-0.0049 ** (0.0025)	-0.0043 * (0.0025)	0.0479 (0.0300)	0.0764 ** (0.0309)	0.0701 ** (0.0309)	-0.0207 *** (0.0024)	0.2360 *** (0.0438)
$[\ln(Cliqness\mathcal{Z}_{t-2} \times 10^4)]^2$	-0.0025 ** (0.0010)	-0.0075 *** (0.0015)				-0.0018 * (0.0010)	-0.0027 *** (0.0010)	-0.0025 ** (0.0010)		-0.0084 *** (0.0015)
Nanotechnology 'career	' age of an aca	ıdemic								
Aget	0.0895 *** (0.0160)	0.0504 ** (0.0184)	0.0312 *** (0.0071)	0.0243 *** (0.0068)	0.0204 ** (0.0066)	0.0689 *** (0.0198)	0.0724 *** (0.0198)	0.0710 *** (0.0197)	-0.0207 *** (0.0074)	-0.0277 (0.0230)
Age_t^2	-0.0027 *** (0.0006)	-0.0018 ** (0.0007)				-0.0020 *** (0.0007)	-0.0022 *** (0.0007)	-0.0021 *** (0.0007)		0.0002 (0.0008)
Interaction variables										
$(Btwness \mathcal{Z}_{t-2} \ge 10^4) \ge 10^{-2}$	-0.0048 ***	-0.0155 ***			-0.0042 ***		-0.0046 ***	-0.0042 ***	-0.0086 **	-0.0113 ***
$\ln(Cliqness \mathcal{J}_{t-2})_{A}$	(0.0010)	(0.0024)			(0.0010)		(0.0011)	(0.0011)	(0.0031)	(0.0025)
(<i>Btwness</i> \mathcal{J}_{t-2} x10 ⁴) x	0.0002	0.0009 **			-0.0001		-0.0002	0.0001	0.0004	0.0006
$\ln(AveCont3_{t-1})$	(0.0002)	(0.0004)			(0.0001)		(0.0002)	(0.0002)	(0.0006)	(0.0004)
(Btwness $3_{t-2} \ge 10^{-1}$) x	-0.0017 ***	-0.0036 ***			-0.0005 ***		-0.0002	-0.0018 ***	-0.0009 ***	-0.0039 ***
$nbPatent3_{t-1}$	(0.0005)	(0.0007)			(0.0001)		(0.0001)	(0.0005)	(0.0001)	(0.0007)

Model	Panel	X-section		Panel Tv	X-section 2SRI					
Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
$(Btwness \beta_{t-2} \times 10^4) \times$	2.6E-05 ***	6.0E-05 ***						2.8E-05 ***		6.4E-05 ***
$[nbPatent3_{t-1}]^2$	(0.9E-05)	(1.3E-05)						(0.9E-05)		(1.3E-05)
Universities dummy vari	ables ^a									
dULaval	-0.2159 ***	-0.0831 *	-0.3256 ***	-0.2949 ***	-0.2565 ***	-0.2707 ***	-0.2391 ***	-0.2359 ***	-0.2441 ***	-0.2130 ***
	(0.0633)	(0.0578)	(0.0729)	(0.0682)	(0.0658)	(0.0659)	(0.6046)	(0.0642)	(0.0621)	(0.0586)
dUMontrealG	-0.2370 ***	-0.1055 **	-0.3458 ***	-0.2922 ***	-0.2653 ***	-0.2798 ***	-0.2488 ***	-0.2491 ***	-0.2269 ***	-0.2091 ***
	(0.0599)	(0.0542)	(0.0677)	(0.0637)	(0.0615)	(0.0616)	(0.0604)	(0.0601)	(0.0578)	(0.0547)
dUConcordia	-0.1412 **	-0.1482 *	-0.0831	-0.1561	-0.1301	-0.1372	-0.1086	-0.1113	-0.0101	-0.0247
	(0.1198)	(0.1340)	(0.1401)	(0.1294)	(0.1241)	(0.1254)	(0.1218)	(0.1214)	(0.1269)	(0.1329)
dUSherbrookeG	-0.3182 ***	-0.1656 ***	-0.4686 ***	-0.4077 ***	-0.3626 ***	-0.3579 ***	-0.3225 ***	-0.3224 ***	-0.2947 ***	-0.2391 ***
	(0.0831)	(0.0702)	(0.0934)	(0.0880)	(0.0849)	(0.0855)	(0.0836)	(0.0833)	(0.0742)	(0.0725)
dUQuebecG	-0.3288 ***	-0.1538 **	-0.4452 ***	-0.3854 ***	-0.3505 ***	-0.3486 ***	-0.3207 ***	-0.3197 ***	-0.1571 **	-0.1227 *
	(0.0728)	(0.0749)	(0.0816)	(0.0768)	(0.0740)	(0.0748)	(0.0732)	(0.0729)	(0.0715)	(0.0676)
Residuals from 1 st stage			-0.0472 **	-0.0516 ***	-0.0448 **	-0.0412 **	-0.0287	-0.0302	-0.2202 ***	-0.2017 ***
_			(0.0190)	(0.0186)	(0.0186)	(0.0192)	(0.0193)	(0.0193)	(0.0346)	(0.0322)
Constant	1.3496 ***	-0.2356 **	1.7311 ***	1.6179 ***	1.6283 ***	1.3036 ***	1.1949 ***	1.2101 ***	-0.8956 ***	-1.6558 ***
	(0.1946)	(0.2295)	(0.1733)	(0.1709)	(0.1694)	(0.2137)	(0.2151)	(0.2148)	(0.2809)	(0.3344)
Years	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
ln(r)	2.6894 ***		2.3980 ***	2.5312 ***	2.6269 ***	2.6322 ***	2.6837 ***	2.6977 ***		
	(0.0837)		(0.0796)	(0.0812)	(0.0824)	(0.0830)	(0.0837)	(0.0840)		
ln(s)	1.5361 ***		1.1760 ***	1.3565 ***	1.4713 ***	1.4539 ***	1.5327 ***	1.5466 ***		
	(0.0799)		(0.0660)	(0.0725)	(0.0766)	(0.0764)	(0.0800)	(0.0805)		
ln(alpha)		-0.9424 ***							-0.9329 ***	-1.0043 ***
		(0.0563)							(0.0521)	(0.0536)
Nb observations	5739	5739	5724	5724	5724	5724	5724	5724	5724	5724
Nb academics	907	907	907	907	907	907	907	907	907	907
Average nb years	6.33	6.33	6.31	6.31	6.31	6.31	6.31	6.31	6.31	6.31
Log likelihood	-11011.2	-11403.2	-11679.5	-11040.6	-11006.8	-10994.9	-10983.3	-10978.3	-11384.3	-11314.1
Wald χ^2	448.85 ***	4985.98 ***	212.25 ***	309.50 ***	390.95 ***	411.84 ***	441.42 ***	455.20 ***	739.13 ***	5651.01 ***

Notes: ***, **, * show significance at the 1%, 5% and 10% levels respectively. Standard errors are presented in parentheses ^a McGill University is the omitted dummy variable

Model	Panel	X-section		Panel Tv		X-section 2SRI				
Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Average amount of oper	ating costs gr	ants over the	past 3 years							
$\ln(AveGrantO3_{t-1})$	-0.0714 *** (0.0154)	-0.1004 *** (0.0201)	0.0599 *** (0.0149)	0.0633 *** (0.0147)	0.0451 *** (0.0135)	-0.0147 (0.0246)	-0.0343 (0.0231)	-0.0331 (0.0231)	0.1550 *** (0.0231)	0.0543 * (0.0319)
$\left[\ln(AveGrantO3_{t-1})\right]^2$	0.0061 *** (0.0013)	0.0094 *** (0.0017)				0.0052 *** (0.0013)	0.0055 *** (0.0013)	0.0055 *** (0.0013)		0.0064 *** (0.0017)
Average amount of cont	racts over the	past 3 years								
$\ln(AveCont\mathcal{Z}_{t-1})$	-0.0090 (0.0136)	-0.0031 (0.0186)	0.0020 (0.0034)	0.0029 (0.0033)	0.0052 (0.0034)	-0.0107 (0.0136)	-0.0105 (0.0136)	-0.0132 (0.0136)	-0.0042 (0.0051)	-0.0173 (0.0181)
$\left[\ln(AveCont\mathcal{Z}_{t-1})\right]^2$	0.0014 (0.0013)	0.0008 (0.0018)				0.0011 (0.0013)	0.0015 (0.0013)	0.0015 (0.0013)		0.0009 (0.0017)
Number of patent applic	cations of the	past 3 years								
$nbPatent3_{t-1}$	0.0789 *** (0.0136)	0.0913 *** (0.0208)	0.0174 *** (0.0044)	-0.0039 (0.0049)	0.0511 *** (0.0084)	0.0696 *** (0.0110)	0.0602 *** (0.0116)	0.0842 *** (0.0137)	0.0535 *** (0.0131)	0.0998 *** (0.0195)
$[nbPatent3_{t-1}]^2$	-0.0012 ** (0.0005)	-0.0020 *** (0.0008)				-0.0016 *** (0.0002)	-0.0007 (0.0005)	-0.0014 ** (0.0005)		-0.0024 *** (0.0008)
3-year network between	ness centrality	,								
$Btwness3_{t-2} \times 10^4$	0.0424 *** (0.0067)	0.1231 *** (0.0146)		0.0102 *** (0.0011)	0.0397 *** (0.0066)	0.0111 *** (0.0011)	0.0412 *** (0.0067)	0.0415 *** (0.0067)	0.0981 *** (0.0177)	0.1094 *** (0.0145)
3-year network cliquish	ness									
$\ln(Cliqness \mathcal{Z}_{t-2} \times 10^4)$	0.0703 ** (0.0309)	0.2080 *** (0.0449)	-0.0044 * (0.0025)	-0.0048 * (0.0025)	-0.0042 * (0.0025)	0.0512 * (0.0300)	0.0789 ** (0.0309)	0.0722 ** (0.0309)	-0.0140 *** (0.0024)	0.2254 *** (0.0448)
$[\ln(Cliqness\mathcal{Z}_{t-2} \times 10^4)]^2$	-0.0025 ** (0.0010)	-0.0075 *** (0.0015)				-0.0019 * (0.0010)	-0.0028 *** (0.0010)	-0.0026 ** (0.0010)		-0.0080 *** (0.0015)
Nanotechnology 'career	' age of an ac	ademic								
Age_t	0.0859 *** (0.0160)	0.0472 ** (0.0184)	0.0267 *** (0.0069)	0.0198 *** (0.0067)	0.0187 *** (0.0064)	0.0599 *** (0.0188)	0.0664 *** (0.0183)	0.0651 *** (0.0183)	-0.0126 * (0.0066)	-0.0060 (0.0206)
Age_t^2	-0.0026 *** (0.0006)	-0.0018 ** (0.0007)				-0.0018 *** (0.0007)	-0.0020 *** (0.0007)	-0.0020 *** (0.0006)		-0.0004 (0.0008)
Interaction variables										
$(Btwness3_{t-2}x10^{4}) x$ $\ln(Cliqness3_{t-2}x10^{4})$	-0.0047 *** (0.0010)	-0.0154 *** (0.0024)			-0.0042 *** (0.0010)		-0.0046 *** (0.0010)	-0.0045 *** (0.0010)	-0.0105 *** (0.0030)	-0.0131 *** (0.0024)
($Btwness3_{t-2}x10^4$) x	0.0002	0.0009 **			-0.0001		-0.0002	0.0002	0.0005	0.0007 *
$ln(AveCont3_{t-1})$ (Btwness3_{t-2}x10 ⁴) x	(0.0002) -0.0017 ***	(0.0004) -0.0036 ***			(0.0001) -0.0005 ***		(0.0002) -0.0002 *	(0.0002) -0.0019 ***	(0.0006) -0.0009 ***	(0.0004) -0.0039 ***
$nbPatent3_{t-1}$	(0.0005)	(0.0007)			(0.0001)		(0.0001)	(0.0005)	(0.0001)	(0.0007)

 Table 2 – Negative binomial regression results – number of articles per academic per year (distinguishing operating cost and infrastructure grants)

Model	Panel	X-section		Panel Tv	vo-Stage Resi	dual Inclusio	on (2SRI)		X-section 2SRI		
Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	
$(Btwness 3_{t-2} \times 10^4) \times$	2.7E-05 ***	6.0E-05 ***						3.0E-05 ***		6.5E-05 ***	
$[nbPatent3_{t-1}]^2$	(0.9E-05)	(1.3E-05)						(0.9E-05)		(1.3E-05)	
Universities dummy var	riables ^a										
dULaval	-0.2125 ***	-0.0769	-0.3303 ***	-0.2984 ***	-0.2534 ***	-0.2702 ***	-0.2364 ***	-0.2325 ***	-0.1854 ***	-0.1522 ***	
	(0.0632)	(0.0575)	(0.0722)	(0.0676)	(0.0652)	(0.0655)	(0.0639)	(0.0636)	(0.0609)	(0.0578)	
dUM on treal G	-0.2311 ***	-0.0956 *	-0.3406 ***	-0.2868 ***	-0.2573 ***	-0.2714 ***	-0.2396 ***	-0.2393 ***	-0.1779 ***	-0.1546 ***	
	(0.0597)	(0.0542)	(0.0674)	(0.0634)	(0.0613)	(0.0614)	(0.0601)	(0.0598)	(0.0562)	(0.0533)	
dUConcordia	-0.1302	-0.1282	-0.0654	-0.1391	-0.1279	-0.1252	-0.0979	-0.1010	-0.0742	-0.0311	
	(0.1196)	(0.1337)	(0.1389)	(0.1282)	(0.1231)	(0.1243)	(0.1207)	(0.1203)	(0.1264)	(0.1335)	
dUSherbrokkeG	-0.3078 ***	-0.1531 **	-0.4759 ***	-0.4139 ***	-0.3633 ***	-0.3560 ***	-0.3166 ***	-0.3156 ***	-0.2617 ***	-0.1997 ***	
	(0.0830)	(0.0705)	(0.0933)	(0.0879)	(0.0849)	(0.0856)	(0.0837)	(0.0833)	(0.0715)	(0.0706)	
dUQuebecG	-0.3186 ***	-0.1396 *	-0.4336 ***	-0.3735 ***	-0.3456 ***	-0.3383 ***	-0.3104 ***	-0.3091 ***	-0.1637 *	-0.1246 *	
	(0.0728)	(0.0748)	(0.0813)	(0.0765)	(0.0739)	(0.0747)	(0.0732)	(0.0728)	(0.0740)	(0.0704)	
Residuals from 1 st stage			-0.0623 ***	-0.0663 ***	-0.0466 ***	-0.0491 ***	-0.0325 **	-0.0336 **	-0.1506 ***	-0.1586 ***	
			(0.0155)	(0.0153)	(0.0142)	(0.0160)	(0.0149)	(0.0148)	(0.0237)	(0.0239)	
Constant	1.3428 ***	-0.2252	1.6480 ***	1.5411 ***	1.6318 ***	1.2716 ***	1.1813 ***	1.2011 ***	-0.3025	-1.0866 ***	
	(0.1941)	(0.2275)	(0.1536)	(0.1516)	(0.1452)	(0.2054)	(0.2056)	(0.2054)	(0.1969)	(0.2863)	
Years	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
1 ()	0 (0 00 state		2 40 52	0.5410	0 (010	0 (07)	a (000	2 7020			
ln(r)	2.6922 ***		2.4052 ***	2.5413 ***	2.6318 ***	2.63/6 ***	2.6888 ***	2.7039 ***			
1 ()	(0.0837)		(0.0797)	(0.0814)	(0.0826)	(0.0831)	(0.0838)	(0.0842)			
ln(s)	1.5479 ***		1.1839 ***	1.3658 ***	1.4/30 ***	1.4617 ***	1.5400 ***	1.5548 ***			
1 (1 1)	(0.0803)	0.0475	(0.0661)	(0.0726)	(0.0765)	(0.0765)	(0.0802)	(0.0807)	0.0107	0.0005	
In(alpha)		-0.94/5 ***							-0.9196 ***	-0.9895 ***	
NH 1		(0.0566)							(0.0533)	(0.0543)	
Nb observations	5741	5741	5710	5710	5/10	5710	5/10	5710	5710	5710	
Nb academics	907	907	906	906	906	906	906	906	906	906	
Average nb years	6.33	6.33	6.30	6.30	6.30	6.30	6.30	6.30	6.30	6.30	
Log likelihood	-11012.1	-11401.1	-11043.6	-11007.8	-10976.1	-10963.4	-10952.7	-10947.1	-11367.5	-11297.9	
Wald χ^2	450.75 ***	5036.74 ***	221.99 ***	397.00 ***	419.87 ***	447.60 ***	447.60 ***	463.05 ***	727.03 ***	5477.35 ***	

Notes: ***, **, * show significance at the 1%, 5% and 10% levels respectively. Standard errors are presented in parentheses ^a McGill University is the omitted dummy variable

Variable	Obs	Mean	Std. Dev.	Min	Max
<i>nbArticle</i> _t	5896	2.6109	2.9233	0	32
$nbPatent3_t$	5896	0.4220	1.7908	0	54
<i>Grant</i> _t	5896	\$260,279	\$1,024,648	0	\$41,900,000
$GrantO_t$	5896	\$184,347	\$328,426	\$84	\$6,738,504
$GrantI_t$	5896	\$75,932	\$910,158	0	\$41,600,000
$Cont_t$	5896	\$45,809	\$392,464	0	\$12,100,000
Ave $Grant3_t$	5896	\$233,503	\$707,174	0	\$15,900,000
$AveGrantO3_t$	5896	\$166,769	\$292,962	\$91	\$5,180,839
Ave $GrantI3_t$	5896	\$66,735	\$553,134	0	\$15,200,000
$AveCont3_t$	5896	\$43,942	\$367,643	0	\$11,600,000
$\ln(AveGrant\mathcal{Z}_t)$	5894	11.4137	1.3499	4.5248	16.5820
$\ln(AveGrantO3_t)$	5896	11.2812	1.2808	4.5248	15.4605
$\ln(AveGrantI3_t)$	5880	3.4135	4.9186	0	16.5350
$\ln(AveCont3_t)$	5891	3.9198	5.0198	0	16.2668
$Btwness3_t$	5896	3.662E-04	9.058E-04	0	2.022E-02
$Cliqness3_t$	5896	0.0157	0.0472	0	0.7243
codeChair	5896	0.6530	0.8845	0	3
Age_t	5896	12.7473	4.3708	1	21

Table B.1 – Descriptive statistics

Table B.2 – Correlation matri

Variable		1	2	3	4	5	6	7	8	9	10 11
<i>nbArticle</i> _t	1	1									
$AveArticle3_{t-2}$	2	0.6283	1								
$nbPatent3_{t-1}$	3	0.1361	0.2340	1							
$ln(AveGrant3_{t-1})$	4	0.0898	0.1499	0.0560	1						
$ln(AveGrantO3_{t-1})$	5	0.0937	0.1550	0.0562	0.9842	1					
$\ln(AveGrantI3_{t-1})$	6	0.0513	0.0201	0.0554	0.3478	0.2743	1				
$\ln(AveCont3_{t-1})$	7	0.0893	0.0890	0.1103	0.2712	0.2701	0.1553	1			
$Btwness3_{t-2} \times 10^4$	8	0.4905	0.7283	0.3123	0.0219	0.0280	-0.0261	0.0109	1		
$\ln(Cliqness 3_{t-2} \times 10^4)$	9	-0.1052	-0.2421	-0.0232	-0.0550	-0.0565	-0.0154	-0.0181	-0.0809	1	
codeChair	10	0.0436	0.0570	-0.0043	0.1735	0.1749	0.1346	0.1178	0.0158	-0.0155	1
Age_t	11	0.1450	0.2598	0.0611	0.3053	0.3140	0.0696	0.1366	0.0937	-0.0756	0.0930 1

TILL C1	T' and a data a				C			
I able C.I -	- First stage	regressions results -	- average amount	of grant	tunding received	l over three vear	's per academi	c ner vear
1	- not stage		a e uge amount	· · · · · · · · · · · · · · · · · · ·		. over enree jear	s per academic	e per jeur

		Panel Two-Stage Residual Inclusion (2SRI)						X-section 2SRI		
Variables	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)		
Instruments										
Average number of articl	les published per	year in the past 3	3-5 years							
AveArticle3 _{t-2}	0.0479 **	0.0740 **	0.0572 *	0.0698 **	0.0491	0.0497	0.1497 ***	0.1673 ***		
	(0.0243)	(0.0309)	(0.0327)	(0.0310)	(0.0330)	(0.0330)	(0.0310)	(0.0308)		
Ordinal variable of the 'b	best' Chair occup	oied by an acaden	nic							
codeChair	0.5703 ***	0.5683 ***	0.5675 ***	0.5712 ***	0.5714 ***	0.5710 ***	0.3970 ***	0.3935 ***		
	(0.0952)	(0.0947)	(0.0947)	(0.0945)	(0.0946)	(0.0947)	(0.0539)	(0.0539)		
Average amount of grant	ts received by nar	notechnology coll	leagues at the san	ne university of a	n academic over	the past 3 years				
ln(AveGrant3U _{t-1})	1.7939 ***	1.7866 ***	1.7769 ***	1.7172 ***	1.7110 ***	1.7101 ***	1.4855 ***	1.4674 ***		
	(0.1124)	(0.1125)	(0.1125)	(0.1121)	(0.1121)	(0.1121)	(0.1758)	(0.1741)		
Variables from the secon	d stage equation									
$\ln(AveCont\mathcal{Z}_{t-1})$	0.0975 ***	0.0971 ***	0.0942 ***	0.0896 **	0.0904 **	0.0921 **	0.1152 ***	0.0967 ***		
	(0.0096)	(0.0096)	(0.0102)	(0.0449)	(0.0449)	(0.0450)	(0.0093)	(0.0408)		
$\left[\ln(AveCont\mathcal{Z}_{t-1})\right]^2$				0.0004	-0.0001	-0.0001		0.0016		
				(0.0042)	(0.0043)	(0.0043)		(0.0038)		
$nbPatent3_{t-1}$	0.0318	0.0394	0.0095	-0.0094	-0.0103	-0.0246	-0.0014	-0.0270		
	(0.0236)	(0.0243)	(0.0328)	(0.0422)	(0.0456)	(0.0509)	(0.0260)	(0.0487)		
$[nbPatent3_{t-1}]^2$				0.0013	0.0010	0.0014		0.0014		
				(0.0010)	(0.0020)	(0.0021)		(0.0016)		
$Btwness \mathcal{J}_{t-2} x 10^4$		-0.0093	0.0324	-0.0152 **	0.0315	0.0314	0.0220	0.0003		
		(0.0069)	(0.0353)	(0.0071)	(0.0358)	(0.0358)	(0.0380)	(0.0384)		
$\ln(Cliqness \mathcal{Z}_{t-2} \times 10^4)$	0.0015	0.0024	0.0021	-0.0525	-0.0274	-0.0235	-0.0002	-0.2591 **		
	(0.0064)	(0.0064)	(0.0064)	(0.0909)	(0.0934)	(0.0936)	(0.0071)	(0.1051)		
$[\ln(Cliqness \beta_{t-2} \times 10^4)]^2$				0.0019	0.0010	0.0009		0.0087 **		
				(0.0031)	(0.0031)	(0.0032)		(0.0036)		
Age_t	0.1775 ***	0.1769 ***	0.1776 ***	0.5630 ***	0.5656 ***	0.5658 ***	0.1351 ***	0.3668 ***		
	(0.0204)	(0.0203)	(0.0204)	(0.0494)	(0.0494)	(0.0494)	(0.0143)	(0.0662)		
Age_t^2				-0.0162 ***	-0.0164 ***	-0.0164 ***		-0.0094 ***		
				(0.0019)	(0.0019)	(0.0019)		(0.0025)		
Interaction variables										
$(Btwness \beta_{t-2} x 10^4) x$			-0.0072		-0.0078	-0.0079	-0.0084	-0.0048		
$\ln(Cliqness \mathcal{J}_{t-2} x 10^4)$			(0.0054)		(0.0055)	(0.0055)	(0.0060)	(0.0061)		

	Panel Two-Stage Residual Inclusion (2SRI)							
Variables	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
(Btwness β_{t-2} x10 ⁴) x			0.0010		0.0012	0.0009	0.0006	0.0005
$\ln(AveCont3_{t-1})$			(0.0008)		(0.0008)	(0.0009)	(0.0006)	(0.0008)
$(Btwness3_{t-2} \times 10^4) \times 10^{-10}$			0.0004		0.0001	0.0016	0.0007 ***	0.0020
$nbPatent3_{t-1}$			(0.0003)		(0.0006)	(0.0025)	(0.0002)	(0.0023)
(Btwness $3_{t-2} \times 10^4$) x					× ,	-2.8E-05	· /	-3.2E-05
$[nbPatent3_{t-1}]^2$						(4.4E-05)		(4.0E-05)
Universities dummy va	riables ^a							
dULaval	-0.1296	-0.1264	-0.1159	-0.1534	-0.1403	-0.1418	0.0303	0.0370
	(0.2501)	(0.2488)	(0.2491)	(0.2483)	(0.2489)	(0.2490)	(0.1517)	(0.1520)
dUMontrealG	-0.2151	-0.2196	-0.2041	-0.2148	-0.2027	-0.2018	0.0306	0.0590
	(0.23433)	(0.2331)	(0.2334)	(0.2328)	(0.2333)	(0.2334)	(0.1453)	(0.1471)
dUConcordia	0.4084	0.4377	0.4667	0.3722	0.3961	0.3858	0.4566	0.4597
	(0.4494)	(0.4479)	(0.4482)	(0.4469)	(0.4475)	(0.4477)	(0.3967)	(0.3951)
dUSherbrookeG	0.2543	0.2542	0.2579	0.2254	0.2346	0.2340	0.3183 *	0.3241 *
	(0.3205)	(0.3188)	(0.3190)	(0.3182)	(0.3182)	(0.3189)	(0.1764)	(0.1783)
dUQuebecG	-0.1840	-0.1890	-0.1784	-0.2777	-0.2664	-0.2663	-0.1812	-0.1842
	(0.2780)	(0.2766)	(0.2770)	(0.2764)	(0.2770)	(0.2771)	(0.1727)	(0.1731)
Constant	-13.1000 ***	-13.0000 ***	-12.9000 ***	-14.1000 ***	-14.1000 ***	-14.2000 ***	-8.6258 ***	-8.5984 ***
	(1.2675)	(1.2684)	(1.2695)	(1.3342)	(1.3360)	(1.3360)	(1.9557)	(1.9991)
Years	yes	yes	yes	yes	yes	yes	yes	yes
Nb observations	5724	5724	5724	5724	5724	5724	5724	5724
Nb academics	907	907	907	907	907	907	907	907
Average nb years	6.31	6.31	6.31	6.31	6.31	6.31	6.31	6.31
F							80.61 ***	364.11 ***
Wald χ^2	2127.23 ***	2127.69 ***	2133,52 ***	2235.94 ***	2241.37 ***	2241.67 ***		
R^2 overall	0.2522	0.2537	0.2547	0.2557	0.2562	0.2563	0.2627	0.2674
R ² within groups	0.2926	0.2921	0.2926	0.3060	0.3067	0.3067		
R ² between groups	0.1301	0.1343	0.1355	0.1251	0.1247	0.1250		

Notes: ***, **, ** show significance at the 1%, 5% and 10% levels respectively. Standard errors are presented in parentheses ^a McGill University is the omitted university

	Panel Two-Stage Residual Inclusion (2SRI)						X-section 2SRI		
Variables	(3)	(4)	(5)	(6)	(7)	(8)	(9)		(10)
Instruments									
Average number of articles p	oublished per yea	r in the past 3-5	years						
$AveArticle3_{t-2}$	0.0440 *	0.0644 **	0.0504	0.0647 **	0.0458	0.0465	0.1405	***	0.1580 ***
	(0.0237)	(0.0302)	(0.0318)	(0.0303)	(0.0321)	(0.0321)	(0.0287)		(0.0289)
Ordinal variable of the 'best	' Chair occupied	by an academic							
codeChair	0.4886 ***	0.4871 ***	0.4785 ***	0.4899 ***	0.4829 ***	0.4825 ***	0.3305	***	0.3272 ***
	(0.0916)	(0.0913)	(0.0915)	(0.0912)	(0.0915)	(0.0915)	(0.0522)		(0.0516)
Average amount of grants re	ceived by nanote	chnology colleag	gues at the same	university of an	academic over	the past 3 years			
$ln(AveGrant3U_{t-1})$	1.6484 ***	1.6425 ***	1.6279 ***	1.5746 ***	1.5631 ***	1.5621 ***	1.2934	***	1.2737 ***
	(0.1104)	(0.1105)	(0.1097)	(0.1100)	(0.1093)	(0.1093)	(0.1709)		(0.1694)
Variables from the second st	age equation								
$ln(AveGrantI3_{t-1})$	0.1164 ***	0.1162 ***	0.1804 ***	0.1948 ***	0.1912 ***	0.1905 ***			-0.0030
	(0.0089)	(0.0089)	(0.0111)	(0.0395)	(0.0393)	(0.0393)			(0.0026)
$\left[\ln(AveGrantI3_{t-1})\right]^2$				-0.0073 **	-0.0013	-0.0013	0.1636	***	0.1959 ***
				(0.0036)	(0.0036)	(0.0036)	(0.0107)		(0.0295)
$ln(AveCont3_{t-1})$	0.0910 ***	0.0907 ***	0.1362 ***	0.0723	0.1065 **	0.1083 **	0.1356	***	0.0947 **
、 · · ·	(0.0094)	(0.0094)	(0.0112)	(0.0440)	(0.0439)	(0.0440)	(0.0109)		(0.0372)
$\left[\ln(AveCont\mathcal{Z}_{t-1})\right]^2$. ,		0.0016	0.0022	0.0022			0.0038
				(0.0041)	(0.0041)	(0.0041)			(0.0035)
$nbPatent3_{t-1}$	0.0359	0.0419 *	0.0030	-0.0411	-0.0333	-0.0483	-0.0193		-0.0771
	(0.0232)	(0.0238)	(0.0320)	(0.0414)	(0.0445)	(0.0497)	(0.0280)		(0.0480)
$[nbPatent3_{t-1}]^2$		× /	× /	0.0022 **	0.0019	0.0024	× /		0.0033 **
				(0.0009)	(0.0019)	(0.0020)			(0.0017)
$Btwness3_{t-2} \times 10^4$		-0.0073	0.0282	-0.0147 **	0.0234	0.0233	0.0057		-0.0124
. 2		(0.0068)	(0.0344)	(0.0070)	(0.0349)	(0.0349)	(0.0360)		(0.0369)
$\ln(Cliqness \beta_{t,2} \times 10^4)$	0.0003	0.0010	-0.0010	-0.0912	-0.0999	-0.0959	-0.0026		-0.2374 **
	(0.0063)	(0.0063)	(0.0063)	(0.0891)	(0.0910)	(0.0912)	(0.0067)		(0.0979)
$\left[\ln(Clianess3_{t-2}\times 10^4)\right]^2$	× /	()	()	0.0031	0.0033	0.0032	× ,		0.0079 **
				(0.0030)	(0.0031)	(0.0031)			(0.0033)
Aget	0.2107 ***	0.2102 ***	0.2151 ***	0.6137 ***	0.6051 ***	0.6053 ***	0.1643	***	0.4125 ***
0-1	(0.0197)	(0.0196)	(0.0197)	(0.0484)	(0.0482)	(0.0482)	(0.0140)		(0.0639)
Age^{2}	(-0.0165 ***	-0.0160 ***	-0.0160 ***			-0.0101 ***
0-1				(0.0018)	(0.0018)	(0.0018)			(0.0024)
Interaction variables									

Table C.2 – First stage regressions results – average amount of operating costs funding received over three years per academic per year

	Panel Two-Stage Residual Inclusion (2SRI)							X-section 2SRI		
Variables	(3)	(4)	(5)	(6)	(7)	(8)	(9)		(10)	
$(Btwness3_{t-2} \times 10^4) \times$			-0.0064		-0.0061	-0.0062	-0.0054		-0.0024	
$\ln(Cliqness \mathcal{Z}_{t-2})$			(0.0053)		(0.0054)	(0.0054)	(0.0057)		(0.0058)	
$\ln(AveGrantI3_{t-2})$ x			-0.0142 ***		-0.0135 ***	-0.0135 ***	-0.0091	***	-0.0090 ***	
$ln(AveCont3_{t-1})$			(0.0015)		(0.0015)	(0.0015)	(0.0012)		(0.0013)	
$(Btwness 3_{t-2} \times 10^4) \times 10^{10}$			0.0009		0.0011	0.0008	0.0005		0.0003	
$\ln(AveCont\mathcal{Z}_{t-1})$			(0.0008)		(0.0008)	(0.0009)	(0.0006)		(0.0007)	
$(Btwness 3_{t-2} \times 10^4) \times 10^{10}$			0.0004		-0.0001	0.0015	0.0008	***	0.0026	
$nbPatent3_{t-1}$			(0.0003)		(0.0006)	(0.0024)	(0.0002)		(0.0022)	
(Btwness $\beta_{t-2} \ge 10^4$) x						-2.9E-05			-4.8E-05	
$[nbPatent3_{t-1}]^2$						(4.3E-05)			(4.0E-05)	
Universitiy dummy variables	a									
dULaval	-0.3757	-0.3730	-0.3812	-0.4153 *	-0.3994 *	-0.4008 *	-0.2601	*	-0.2558 *	
	(0.2409)	(0.2400)	(0.2407)	(0.2399)	(0.2408)	(0.2408)	(0.1516)		(0.1526)	
dUMontrealG	-0.4244 *	-0.4276 *	-0.4515 **	-0.4190 *	-0.4418 *	-0.4409 *	-0.1840		-0.1484	
	(0.2256)	(0.2248)	(0.2254)	(0.2246)	(0.2254)	(0.2254)	(0.1458)		(0.1473)	
dUConcordia	0.3623	0.3848	0.4396	0.3293	0.3812	0.3810	0.2880		0.2903	
	(0.4332)	(0.4324)	(0.4328)	(0.4317)	(0.4323)	(0.4323)	(0.3588)		(0.3511)	
dUSherbrookeG	-0.0698	-0.0700	-0.0761	-0.1224	-0.0973	-0.0978	-0.0874		-0.0831	
	(0.3086)	(0.3074)	(0.3082)	(0.3074)	(0.3084)	(0.3083)	(0.1716)		(0.1731)	
dUQuebecG	-0.3518	-0.3559	-0.3867	-0.4541 *	-0.4617 *	-0.4616 *	-0.4447	***	-0.4482 ***	
-	(0.2673)	(0.2663)	(0.2672)	(0.2665)	(0.2674)	(0.2674)	(0.1562)		(0.1568)	
Constant	-11.8000 ***	-11.7000 ***	-11.7000 ***	-12.7000 ***	-12.6000 ***	-12.6000 ***	-6.8274	***	-6.9823 ***	
	(1.2449)	(1.2459)	(1.2375)	(1.3097)	(1.3030)	(1.3031)	(1.8960)		(1.9309)	
Years	yes	yes	yes	yes	yes	yes	yes		yes	
Nb observations	5710	5710	5710	5710	5710	5710	5710		5710	
Nb academics	906	906	906	906	906	906	906		906	
Average nb years	30	30	30	30	30	30	30		30	
F							95.27	***	452.41 ***	
Wald c^2	2308.71 ***	2308.92 ***	2443.96 ***	2442.29 ***	2560.28 ***	2560.46 ***				
R ² overall	0.2792	0.2802	0.2850	0.2848	0.2885	0.2886	0.2947		0.3003	
R ² within groups	0.3047	0.3062	0.3205	0.3216	0.3341	0.3341				
R ² between groups	0.1700	0.1737	0.1744	0.1668	0.1657	0.1661				

Notes: ***, **, * show significance at the 1%, 5% and 10% levels respectively. Standard errors are presented in parentheses ^a McGill University is the omitted university